Author's response to reviews

Title: Acquired A amyloidosis from injection drug use presenting with atraumatic splenic rupture in a hospitalized patient: a case report

Authors:

Garrett R Roll (Garrett.Roll@ucsfmedctr.org)
Andrew Y Lee (LeeAY2@stemcell.ucsf.edu)
Kayvan Royaie (kayvanr@yahoo.com)
Brendan Visser (bvisser@stanford.edu)
Douglas K Hanks (Dhanks7960@aol.com)
Margaret M Knudson (pknudson@sfghsurg.ucsf.edu)
Frederick J Roll (jroll@itsa.ucsf.edu)

Version: 3 Date: 20 July 2010

Author's response to reviews: see over
Dear JCMR editors,

Thank you very much for your consideration of MS: 1752804568368604, entitled: Acquired Amyloidosis from injection drug use presenting with atraumatic splenic rupture in a hospitalized patient: a case report. We appreciate the editorial team’s comments and concerns, and have addressed them accordingly.

We confirm that the following three conditions have been met with regards to patient consent in our manuscript:

1. All reasonable attempts to gain consent have been made from the patient or their next-of-kin
2. Every effort has been made to protect the identity of the patient
3. There is no reason to think that the patient or their family would object to publication

As such, we have changed the wording of the Consent section as recommended to read as follows:

“Written informed consent for publication could not be obtained despite all reasonable attempts. Every effort has been made to protect the identity of the patient and there is no reason to believe that the patient would object to publication.”

Once again, thank you very much for your consideration of our manuscript.

Garrett R. Roll, MD
General Surgery Resident and Hepatology Research Fellow
513 Parnassus, Room S-321
University of California San Francisco
San Francisco, CA 94143
Garrett.Roll@ucsfmedctr.org
(W) 415-476-2420
(C) 415-420-7144